1

2

3

4 5 6

7

manuscript submitted to Geophysical Research Letters

Rapid cooling and increased storminess triggered by freshwater in the North Atlantic

M. Oltmanns¹, J. Karstensen², G. W. K. Moore³, S. A. Josey¹

¹National Oceanography Centre, Southampton, UK
²GEOMAR - Helmholtz Centre for Ocean Research Kiel, Kiel, Germany
³University of Toronto, Toronto, Canada

Key Points:

| 8 | • | Large freshwater events result in distinct cold anomalies with sharp temperature |
|----|---|-----------------------------------------------------------------------------------|
| 9 | | gradients in the subpolar North Atlantic in winter. |
| 10 | • | A strong, freshwater-induced cold anomaly promotes an enhanced storminess, which |
| 11 | | reinforces the anomaly by modulating the surface flow. |
| 12 | • | Consistent with this mechanism, large freshwater events in the past have preceded |
| 13 | | positive North Atlantic Oscillation periods in winter. |

Corresponding author: Marilena Oltmanns, marilena.oltmanns@noc.ac.uk

14 Abstract

- Recent winters have been unique due to the rapid and extreme cooling of the subpolar
- ¹⁶ North Atlantic. Here, we present a novel view on its causes and consequences. Combin-
- ¹⁷ ing in-situ observations with remote sensing and atmospheric reanalysis data, we show

that increased freshening of the subpolar region gives rise to a faster surface cooling in

- fall and winter. Large freshwater events, in particular, result in pronounced cold anomalies with sharp temperature gradients that promote an enhanced storminess. The storms
- reinforce the cooling by driving stronger heat losses and modulating the surface flow. Con-
- sistent with this mechanism, past freshwater events have been followed by cold anoma-
- lies in winter of ~ -2 °C and increases in the North Atlantic Oscillation index of up to
- ~ 0.6 within 3 years. We expect that future freshwater discharges into the North Atlantic
- ²⁵ will amplify the cold anomaly and trigger an enhanced wintertime storminess with far-
- ²⁶ reaching climatic implications.

27 Plain Language Summary

Recent winters have been unique due to a rapid and extreme cooling of the sub-28 polar North Atlantic. Combining ocean and atmospheric data, we show that increased 29 freshwater in this region leads to shallower surface layers that adjust faster to the lower 30 air temperature in fall and winter. The faster surface cooling increases the south-north 31 temperature gradient which promotes the development of storms. The storms, in turn, 32 33 reinforce the cooling by modulating the surface flow. Accordingly, past freshwater events have been followed by an extremely cold ocean surface in the subpolar North Atlanic in 34 winter and major changes in large-scale weather patterns. We expect that future fresh-35 water discharges from Greenland and the Arctic will amplify the cooling and trigger an 36 enhanced wintertime storminess with far-reaching implications for the climate. 37

38 1 Introduction

Recent winters have been characterised by a rapid and extreme cooling of the subpolar North Atlantic, which has been unprecedented in over 30 years and stands in marked contrast to the warming observed over most of the Earth's surface (Josey et al., 2018). Given the importance of the North Atlantic sea surface temperature (SST) for the largescale weather and climate (Czaja & Frankignoul, 2002; Sutton & Dong, 2012), it is critical to understand the causes and effects of this anomaly.

Previous studies have attributed the cooling to a slowdown of the Atlantic over-45 turning circulation due to increased freshwater fluxes from Greenland (Rahmstorf et al., 46 2015; Caesar et al., 2018). This idea is motivated by paleoclimate records, suggesting 47 that past cooling events were caused by a freshwater-forced shutdown of deep ocean con-48 vection in the subpolar North Atlantic and a subsequent collapse of the overturning (Barber 49 et al., 1999; Clark et al., 2001, 2002). However, observations show that this buoyancy-50 driven mechanism cannot easily explain the recent heat transport into the subpolar re-51 gion (Lozier et al., 2019). 52

The influences of the cold anomaly on the climate are likewise uncertain. Yet, earlier studies have found that *increased* SSTs in the subpolar region initially trigger a transient baroclinic response in the atmosphere, forcing enhanced ocean heat losses (Kushnir et al., 2002; Deser et al., 2007). After a few weeks, a barotropic equilibrium response emerges that is associated with reduced ocean heat losses and thus represents a positive feedback to the SST anomaly (Kushnir et al., 2002; Deser et al., 2007).

However, we hypothesise that freshwater modulates this atmospheric response by
 strengthening the stratification, allowing for a faster adjustment of the surface to the lower
 air temperatures in fall and winter. By eroding the SST anomaly during the transient

| Data/index | Period |
|--------------------------------------------------|-------------|
| Hydrographic observations from Argo floats | 2002-2018 |
| Remote sensing-based SST data | 1982 - 2018 |
| Absolute dynamic topography from altimetry | 1993 - 2018 |
| Atmospheric data from the reanalysis ERA-Interim | 1979 - 2018 |
| Freshwater index $(\mathbf{F}_{NA})^a$ | 1979 - 2018 |
| Cold anomaly index $(CAI)^b$ | 1982 - 2019 |

 Table 1. Main data products and indices involved in this study.

^{*a*} F_{NA} : Mean NAO in July and August, multiplied by -1.

^b CAI: Mean SST in the cold anomaly region, multiplied by -1.

response, the faster surface cooling interferes with the setup of the equilibrium response and instead, increases the meridional SST gradient, key source of baroclinic instability (Hoskins et al., 1985). To test this hypothesis, we investigate the chain of mechanisms initiated by freshwater. The main data products involved in this study are listed in Table 1, while a detailed data description is provided in the Supporting Information.

67 2 Results

68

85

2.1 Detection of surface freshwater

Motivated by earlier studies that discovered a significant anti-correlation of the North Atlantic Oscillation (NAO) in summer with the melting over Greenland (Hanna et al., 2013) and the Arctic sea ice export (Haine et al., 2015), both potential sources of freshwater, we use the mean NAO in July and August, multiplied by -1, as an index ('F_{NA}') to describe the freshwater variability in the subpolar North Atlantic (Fig. 1a).

⁷⁴ Based on a scaling analysis of the surface mass balance, we find that the surface ⁷⁵ freshening associated with F_{NA} can be estimated from:

$$\beta \Delta SSS \approx \alpha \Delta SST,\tag{1}$$

where both sides have been regressed on F_{NA} , α and β are the thermal and haline expansion coefficients, SSS is the surface salinity and Δ refers to the change from summer to winter (see Supporting Information for details).

The inferred surface freshening is most pronounced in the western subpolar region, off the coast of Newfoundland, from where it expands eastward into the central gyre, following the mean geostrophic flow (Fig. 1b). Both F_{NA} and the inferred freshening are characterised by a significantly positive trend over the investigated period and their correlation is largest on time scales above 5 years (Fig. 1c). Thus, F_{NA} identifies periods of increased freshwater rather than individual years.

2.2 Influence of surface freshwater on the SST

The surface mass balance states that an enhanced surface freshening strengthens the stratification, which allows for an increased surface cooling in fall and winter before the freshwater is mixed down. This freshwater-induced surface cooling is one order of magnitude larger than the cooling resulting from the anomalous surface heat flux associated with F_{NA} .

⁹¹ Hydrographic observations from the Labrador Sea over the period 2002–2018 show ⁹² that an increased surface freshening not only leads to an anomalous cooling relative to



Figure 1. (a) Variability of F_{NA} , which is the mean NAO index in July and August, multiplied by -1, and the inferred surface freshening $(-\Delta SSS)$ from summer (August) to winter (January through March) in the region enclosed by the blue 95% confidence line in b. Also shown are the trends. (b) Regression of the inferred SSS change on F_{NA} in the region where an anomalous surface cooling is observed. Thick contours mark the 95% confidence level and arrows indicate the direction of the mean geostrophic flow. (c) Multi-tapered coherence and phase shift between F_{NA} and the seasonal SSS change in the region enclosed by the blue 95% confidence line in b, computed using 8 tapers. Significance analyses are provided in the Supporting Information.



Figure 2. Regressions of (a) the SST change from summer (August) to winter (January through March) and (b,c) the absolute salinity and potential temperature in the Labrador Sea on F_{NA} , based on 17-year long mooring and Argo float observations. The black contour in a delineates the Argo float sampling region and the circle marks the mooring location. Coloured contours indicate the 95% confidence levels.

the summer (Fig. 2a) but also to a cold surface anomaly relative to the climatological mean (Fig. 2b and c). While the positive subsurface temperature and salinity anomalies are expected from the increased advection of heat and salt in negative NAO periods (Sarafanov, 2009; Reverdin, 2010), the cold surface anomaly can only be explained by the freshwater-induced surface cooling.

However, the hydrographic observations were acquired over a time with an overall elevated freshening (Fig. 1a). When considering the full period of satellite observations, the anti-correlation between the freshening in summer and the SST in the subsequent winter only holds for positive freshwater anomalies (Fig. 3a-c), reflecting the transition from the regime where the surface heat fluxes control the SST anomaly to the regime where the freshwater dominates.

The freshwater-induced cold anomaly is most pronounced in the eastern subpolar region (Fig. 3a and b), where the correlation of the SST, multiplied by -1 (the cold anomaly index 'CAI'), with the positive F_{NA} values amounts to ~0.57, which increases for higher values of F_{NA} (Fig. 3c). The associated heat flux anomaly is directed upwards and thus cannot account for the cold anomaly. On the contrary, the cold anomaly weakens the heat loss to the atmosphere by reducing the air-sea temperature contrast (Fig. 3d).

Southeast of the cold anomaly over the Gulf Stream, the SST is increased (Fig. 3a), giving rise to a sharper meriodonal SST gradient, key source of baroclinic instability in the atmosphere (Hoskins et al., 1985). Thus, we next investigate the interaction of the cold anomaly with the atmosphere.

2.3 Atmospheric feedbacks

114

Regressing the sea level pressure (SLP) onto the cold anomaly, we find that the cold anomaly is accompanied by an anomalous low over and southeast of Greenland (Fig. 4a), representative of a positive NAO. In addition, the standard deviation of the 2-6 day bandpass filtered SLP is increased (Fig. 4b), implying an enhanced storminess (Blackmon, 1976; Blackmon et al., 1977; Ulbrich et al., 2008). Consistent with the increased cyclonic activity, the heat loss over the Labrador Sea, south of the sea ice edge, is amplified (Fig. 4c), and the wind stress curl under the low is higher (Fig. 4d).



Figure 3. (a,b,d) Regression of the SST, the inferred SSS in the cold anomaly region, and the surface heat fluxes in winter (January through March) on F_{NA} from the preceding summer, with only positive F_{NA} included. A positive heat flux anomaly means that the ocean loses less heat. Thick contours indicate the 95% confidence levels. (c) Relationship between F_{NA} and the SST in the region delineated by the thick blue contour in a. When multiplied by -1, this SST anomaly corresponds to the cold anomaly index (CAI). Also shown are the correlation coefficients and regression lines for $F_{NA} > 0$ and $F_{NA} > 0.5$.



Figure 4. (a–d) Regressions of (a) the SLP, (b) the 2 to 6-day band-pass filtered standard deviation of the SLP, (c) the surface heat flux (positive downward) and (d) the wind stress curl on CAI, multiplied by the regression of CAI on positive F_{NA} . (e) Regressions of the absolute dynamic topography on the wind stress curl within the red contour in d ('WSC'), multiplied by the regressions of WSC on CAI and CAI on positive F_{NA} . (f) Correlations of CAI with F_{NA} and the WSC, with the envelopes indicating the 95% confidence levels. A positive lag implies that the cold anomaly lags F_{NA} and the WSC. All variables are in winter (January through March).

A high wind stress curl has been identified as an important driver of the subpolar gyre circulation, which retains cold, fresh polar water in the interior gyre (Häkkinen et al., 2011; Chafik et al., 2019). Altimetry data confirm that the obtained wind stress curl is associated with a stronger cyclonic circulation, implying an increased advection of polar water off the coast of Newfoundland towards the anomaly, hence reinforcing it (Fig. 4e). The correlation between the surface flow pattern and the cold anomaly is ~0.80.

¹²⁸ 2.4 Summary of the feedback mechanism

¹²⁹ Closer inspection of the timing of the correlations reveals that the cold anomaly ¹³⁰ emerges in the winter immediately *following* a high F_{NA} summer and occurs in *the same* ¹³¹ winter with the enhanced wind stress curl (Fig. 4f). It then continues to intensify and ¹³² peaks one year *after* the wind stress curl reaches its maximum (Fig. 4f), as expected from ¹³³ the subpolar gyre response (Lohmann et al., 2009). We sum up:

- 1. Increased freshening of the subpolar region gives rise to a faster surface cooling.
 - 2. After large freshwater anomalies, a distinct cold anomaly appears.
- ¹³⁶ 3. The cold anomaly promotes an enhanced storminess.
- 4. The resulting higher wind stress curl strengthens the subpolar gyre circulation.
- 5. The stronger subpolar gyre circulation reinforces the cold anomaly.

This chain of events is supported by the change in the sign of the heat flux anomaly, which is initially downward due to the reduced air-sea contrast (Fig. 3d) and then upward and driven by the atmospheric forcing (Fig. 4c). However, the discrepancy between the two heat flux anomalies also reveals a nonlinearity in the direct relation between freshwater and the atmosphere: There exists a threshold for the freshwater, after which the sign of the heat flux anomaly reverses, reflecting the transition from the ocean-driven to the atmospherically-driven heat flux anomaly.

A composite of the largest freshwater events over the investigated period, included in the Supporting Information, shows the transition of the heat flux anomaly from being positive in January and February to negative in March. It shows that the most *negative* NAO summers were followed by a *positive* NAO in winter. A regression on these large freshwater events reveals, in addition, a high sensitivity of the atmospheric response to small variations in the freshwater when the freshwater concentration is already high, consistent with the nonlinear relation between F_{NA} and the SST (Figs. 3c and S7–S10).

153

135

2.5 Implications of freshwater events for the North Atlantic climate

Considering that the observational record is still short with regard to the occur-154 rence of large freshwater events, we inspect the evolution of these events individually. 155 Two large freshwater events are the Great Salinity Anomaly in 1969–1972 (Lazier, 1980) 156 and the recent freshening (Holliday et al., 2020) peaking in 2008–2012 and 2015/2016 157 (Supporting Information). The Great Salinity Anomaly preceded a cold anomaly of \sim -158 1.5 °C while the recent freshening culminated in an anomalous ~ -2 °C (Supporting In-159 formation). Both events were followed by an enhanced storminess, reflected in increases 160 of the winter NAO of ~ 0.61 after 1970 and ~ 0.72 after 2010 (Fig. 5a). 161

Additional salinity anomalies have been reported for the 1980's and 1990's (Belkin et al., 1998). Although the sparse sampling prevents the determination of the exact starting dates, compilations of historical salinity data suggest that they appeared in the Labrador Sea in 1980 (Reverdin et al., 1997; Yashayaev & Loder, 2016) and 1993 (Yashayaev & Clarke, 2006; Yashayaev & Loder, 2016), consistent with increased values of F_{NA} (Fig. 1a). Both events were followed by a reduced SLP and a negative SST anomaly in the subpolar region within three years (Fig. 5).



Figure 5. (a) Variability of the 3-winter low-pass filtered NAO index. The solid lines and shadings indicate the times of large documented freshwater anomalies and the dashed line marks the time of an inferred event. (b-g) SLP and SST anomalies following the events in 1980, 1986 and 1993. While the anomalies typically last for several winters, we show them at the time when they reach their maximum amplitude: (b) SLP 1982, (c) SST 1983, (d) SLP 1989, (e) SST 1990, (f) SLP 1994, (g) SST 1994. All anomalies are with respect to the climatological winter mean.

The detailed evolution of the cold anomaly differed between the investigated freshwater events, which we attribute to variations in the freshwater volumes, their pathways (Belkin et al., 1998) and the strength of the surface fluxes, mixing freshwater down. Also, the event in 1980 preceded an increase in the winter NAO of ~0.46, similar to the other events, whereas the event in 1993 occurred in a phase, in which the winter NAO was already high (Fig. 5a). Thus, the subpolar gyre circulation was already intensified (Belkin, 2004) and the maximum SST anomaly was reached sooner (Fig. 5g).

Examining the winter NAO more closely, we detect another period of increase starting in 1986 (Fig. 5a), concurrent with an elevated value of F_{NA} (Fig. 1a). The following winters were characterised by a strongly reduced SLP and a distinct cold anomaly, exceeding ~-2 °C in the Labrador Sea (Fig. 5d and e). We conclude that, despite differences in their amplitude and detailed evolution, all freshwater events show a link with an enhanced wintertime storminess and the emergence of a cold anomaly. Conversely, all major NAO increases in the last 50 years show a link with a freshwater event.

183 **3** Conclusion

Combining in-situ observations with remote sensing and atmospheric reanalysis data, we have shown that enhanced freshening of the subpolar North Atlantic gives rise to an increased surface cooling in fall and winter. Over the last four decades, the freshwaterinduced surface cooling has been characterised by a significantly positive trend, reflecting a growing discrepancy between the summer and winter SSTs.

¹⁸⁹ Superimposed on the trend, individual strong freshwater events have triggered pro-¹⁹⁰ nounced cold anomalies in winter. By increasing the meridional SST gradient, cold anoma-¹⁹¹ lies in the subpolar region promote the development of cyclones, which reinforce the cool-¹⁹² ing by modulating the subolar gyre circulation. In agreement with this chain of events, ¹⁹³ past freshwater events were followed by cold anomalies of \sim -2 °C and an enhanced win-¹⁹⁴ tertime storminess, reflected in increases of the NAO of up to \sim 0.6 within three years.

Our findings suggest that the cold anomaly in winter can be explained by an enhanced surface freshening and the resulting regional atmosphere-ocean interactions, without the need for a buoyancy-driven slowdown of the large-scale overturning circulation. Thus, this study reconciles the proposed inconsistencies between the recent strong convection in the Labrador Sea and the reduced northward heat transport (Lozier et al., 2019). However, the results do not exclude the possibility of a buoyancy-driven slowdown of the overturning on longer time scales.

The identified, freshwater-induced SST pattern resembles the negative phase of the Atlantic Multidecadal Oscillation, which has been found to drive a positive NAO response in winter (Peings & Magnusdottir, 2014; Gastineau & Frankignoul, 2015). Our findings are therefore consistent with earlier studies and suggest that freshwater plays a key role in modulating the low-frequency climate variability over the North Atlantic. In particular, they show that large freshwater events, arising from the most negative NAO summers, were followed by rapid transitions to the most positive NAO phases in winter.

With regard to the influences of the winter NAO on large-scale weather patterns 209 (Hurrell & Deser, 2010), freshwater-induced NAO increases have far-reaching implica-210 tions. For instance, they can trigger cold spells over Canada (Shabbar & Bonsal, 2004), 211 precipitation events over the UK (Lavers et al., 2011) and storms over northwest Europe 212 (Feser et al., 2015). Considering the growing freshwater discharges from Greenland (Bamber 213 et al., 2012) and the Arctic (Haine et al., 2015), and the high sensitivity to small vari-214 ations in the freshwater, when the freshwater concentration is already high, our results 215 raise caution that weather extremes associated with deep cyclones in the subpolar re-216 gion in winter will increase. 217

218 Acknowledgments

The Argo data were collected and made freely available by the international Argo project and the national programs that contribute to it (http://doi.org/10.17882/42182). We further thank the National Oceanic and Atmospheric Administration and the Hadley Centre for providing the SST data (https://psl.noaa.gov/data/gridded/data.noaa

- .oisst.v2.html and https://www.metoffice.gov.uk/hadobs/hadisst/), the Coper-
- nicus Marine Service for distributing the altimeter products (https://marine.copernicus
- .eu) and the European Centre for Medium-Range Weather Forecasts for their reanal-
- ysis products (https://www.ecmwf.int/en/forecasts/datasets).

This study was supported by the EU Horizon 2020 programme under grant agreements 727852 and 633211, the German BMBF project "RACE" and the UK NERC programmes NE/N018044/1 and NE/R015953/1.

230 References

250

251

252

253

254

255

264

265

- Bamber, J., van den Broeke, M., Ettema, J., Lenaerts, J., & Rignot, E. (2012).
 Recent large increases in freshwater fluxes from Greenland into the North
 Atlantic. *Geophysical Research Letters*, 39(19).
- Barber, D. C., Dyke, A., Hillaire-Marcel, C., Jennings, A. E., Andrews, J. T., Kerwin, M. W., ... others (1999). Forcing of the cold event of 8,200 years ago by
 catastrophic drainage of Laurentide lakes. *Nature*, 400(6742), 344.
- Barnston, A. G., & Livezey, R. E. (1987). Classification, seasonality and persistence of low-frequency atmospheric circulation patterns. *Monthly weather review*, 115(6), 1083–1126.
- Belkin, I. M. (2004). Propagation of the "great salinity anomaly" of the 1990s
 around the northern north atlantic. *Geophysical Research Letters*, 31(8).
- Belkin, I. M., Levitus, S., Antonov, J., & Malmberg, S.-A. (1998). "great salinity anomalies" in the North Atlantic. *Progress in Oceanography*, 41(1), 1–68.
- Blackmon, M. L. (1976). A climatological spectral study of the 500 mb geopoten tial height of the northern hemisphere. Journal of the Atmospheric Sciences,
 33(8), 1607–1623.
- Blackmon, M. L., Wallace, J. M., Lau, N.-C., & Mullen, S. L. (1977). An observational study of the Northern Hemisphere wintertime circulation. Journal of the Atmospheric Sciences, 34(7), 1040–1053.
 - Caesar, L., Rahmstorf, S., Robinson, A., Feulner, G., & Saba, V. (2018). Observed fingerprint of a weakening atlantic ocean overturning circulation. Nature, 556 (7700), 191.
 - Chafik, L., Nilsen, J. E. Ø., Dangendorf, S., Reverdin, G., & Frederikse, T. (2019). North atlantic ocean circulation and decadal sea level change during the altimetry era. *Scientific reports*, 9(1), 1–9.
- Clark, P. U., Marshall, S. J., Clarke, G. K., Hostetler, S. W., Licciardi, J. M., &
 Teller, J. T. (2001). Freshwater forcing of abrupt climate change during the
 last glaciation. *Science*, 293(5528), 283–287.
- Clark, P. U., Pisias, N. G., Stocker, T. F., & Weaver, A. J. (2002). The role of the
 thermohaline circulation in abrupt climate change. *Nature*, 415 (6874), 863.
- Cronin, M., & Sprintall, J. (2009). Wind-and buoyancy-forced upper ocean. El *ements of Physical Oceanography: A derivative of the Encyclopedia of Ocean* Sciences, 237–245.
 - Czaja, A., & Frankignoul, C. (2002). Observed impact of atlantic sst anomalies on the North Atlantic Oscillation. *Journal of Climate*, 15(6), 606–623.
- Dee, D. P., Uppala, S., Simmons, A., Berrisford, P., Poli, P., Kobayashi, S., ... oth ers (2011). The era-interim reanalysis: Configuration and performance of the
 data assimilation system. *Quarterly Journal of the royal meteorological society*,
 137(656), 553–597.

- Deser, C., Tomas, R. A., & Peng, S. (2007).The transient atmospheric circula-270 tion response to north atlantic sst and sea ice anomalies. Journal of Climate, 271 20(18), 4751-4767.272 Feser, F., Barcikowska, M., Krueger, O., Schenk, F., Weisse, R., & Xia, L. (2015).273 Storminess over the north atlantic and northwestern europe—a review. Quar-274 terly Journal of the Royal Meteorological Society, 141(687), 350–382. 275
- Foukal, N. P., & Lozier, M. S. (2018). Examining the origins of ocean heat content
 variability in the eastern north atlantic subpolar gyre. *Geophysical Research Letters*, 45(20), 11–275.
- Gastineau, G., & Frankignoul, C. (2015). Influence of the north atlantic sst variabil ity on the atmospheric circulation during the twentieth century. Journal of Cli mate, 28(4), 1396-1416.
- 282 Gill, A. E. (2016). Atmosphere—ocean dynamics. Elsevier.

300

301

305

306

307

308

309

- Griffies, S. M., & Greatbatch, R. J. (2012). Physical processes that impact the evolution of global mean sea level in ocean climate models. Ocean Modelling, 51, 37–72.
- Haine, T. W., Curry, B., Gerdes, R., Hansen, E., Karcher, M., Lee, C., ... others
 (2015). Arctic freshwater export: Status, mechanisms, and prospects. *Global* and Planetary Change, 125, 13–35.
- Häkkinen, S., Rhines, P. B., & Worthen, D. L. (2011). Warm and saline events embedded in the meridional circulation of the northern north atlantic. Journal of Geophysical Research: Oceans, 116(C3).
- Hanna, E., Jones, J. M., Cappelen, J., Mernild, S. H., Wood, L., Steffen, K., & Huybrechts, P. (2013). The influence of north atlantic atmospheric and oceanic
 forcing effects on 1900–2010 greenland summer climate and ice melt/runoff. *International Journal of Climatology*, 33(4), 862–880.
- Holliday, N. P., Bersch, M., Berx, B., Chafik, L., Cunningham, S., Florindo-López,
 C., ... others (2020). Ocean circulation causes the largest freshening event for
 120 years in eastern subpolar north atlantic. Nature Communications, 11(1),
 1–15.
 - Holte, J., Talley, L. D., Gilson, J., & Roemmich, D. (2017). An argo mixed layer climatology and database. *Geophysical Research Letters*, 44 (11), 5618–5626.
- Hoskins, B. J., McIntyre, M. E., & Robertson, A. W. (1985). On the use and significance of isentropic potential vorticity maps. *Quarterly Journal of the Royal Meteorological Society*, 111(470), 877–946.
 - Hurrell, J. W., & Deser, C. (2010). North atlantic climate variability: the role of the north atlantic oscillation. Journal of Marine Systems, 79(3-4), 231–244.
 - Josey, S. A., Hirschi, J. J.-M., Sinha, B., Duchez, A., Grist, J. P., & Marsh, R.
 - (2018). The recent atlantic cold anomaly: Causes, consequences, and related phenomena. Annual review of marine science, 10, 475–501.
- Kennedy, J., Rayner, N., Smith, R., Parker, D., & Saunby, M. (2011). Reassessing
 biases and other uncertainties in sea surface temperature observations measured in situ since 1850: 1. measurement and sampling uncertainties. Journal
 of Geophysical Research: Atmospheres, 116(D14).
- Kushnir, Y., Robinson, W., Bladé, I., Hall, N., Peng, S., & Sutton, R. (2002). Atmospheric gcm response to extratropical sst anomalies: Synthesis and evaluation.
 Journal of Climate, 15(16), 2233–2256.
- Lavers, D. A., Allan, R. P., Wood, E. F., Villarini, G., Brayshaw, D. J., & Wade,
 A. J. (2011). Winter floods in britain are connected to atmospheric rivers. *Geophysical Research Letters*, 38(23).
- Lazier, J. R. (1980). Oceanographic conditions at ocean weather ship bravo, 1964– 1974. Atmosphere-ocean, 18(3), 227–238.
- Le Traon, P., Nadal, F., & Ducet, N. (1998). An improved mapping method of multisatellite altimeter data. *Journal of atmospheric and oceanic technology*, 15(2), 522–534.

Lohmann, K., Drange, H., & Bentsen, M. (2009).Response of the north atlantic 325 subpolar gyre to persistent north atlantic oscillation like forcing. Climate dy-326 namics, 32(2-3), 273–285. 327 Lozier, M., Li, F., Bacon, S., Bahr, F., Bower, A., Cunningham, S., ... others 328 (2019). A sea change in our view of overturning in the subpolar north atlantic. 329 Science, 363(6426), 516–521. 330 Peings, Y., & Magnusdottir, G. (2014). Forcing of the wintertime atmospheric cir-331 culation by the multidecadal fluctuations of the north atlantic ocean. Environ-332 mental Research Letters, 9(3), 034018. 333 Percival, D. B., Walden, A. T., et al. (1993). Spectral analysis for physical applica-334 tions. cambridge university press. 335 Rahmstorf, S., Box, J. E., Feulner, G., Mann, M. E., Robinson, A., Rutherford, S., 336 & Schaffernicht, E. J. (2015).Exceptional twentieth-century slowdown in 337 atlantic ocean overturning circulation. Nature climate change, 5(5), 475. 338 339 Reverdin, G. (2010). North atlantic subpolar gyre surface variability (1895–2009). Journal of climate, 23(17), 4571–4584. 340 Reverdin, G., Cayan, D., & Kushnir, Y. (1997). Decadal variability of hydrography 341 in the upper northern north atlantic in 1948–1990. Journal of Geophysical Re-342 search: Oceans, 102(C4), 8505-8531. 343 Reynolds, R. W., Ravner, N. A., Smith, T. M., Stokes, D. C., & Wang, W. (2002).344 An improved in situ and satellite sst analysis for climate. Journal of climate, 345 15(13), 1609-1625.346 Sarafanov, A. (2009). On the effect of the north atlantic oscillation on temperature 347 and salinity of the subpolar north atlantic intermediate and deep waters. *ICES* 348 Journal of Marine Science, 66(7), 1448–1454. 349 Shabbar, A., & Bonsal, B. (2004). Associations between low frequency variability 350 modes and winter temperature extremes in canada. Atmosphere-Ocean, 42(2), 351 127 - 140.352 (2003).Wind-driven recirculations and exchange Spall, M. A., & Pickart, R. S. 353 in the labrador and irminger seas. Journal of Physical Oceanography, 33(8), 354 1829-1845. 355 Sutton, R. T., & Dong, B. (2012). Atlantic ocean influence on a shift in european 356 climate in the 1990s. Nature Geoscience, 5(11), 788. 357 Ulbrich, U., Pinto, J. G., Kupfer, H., Leckebusch, G., Spangehl, T., & Reyers, M. 358 (2008).Changing northern hemisphere storm tracks in an ensemble of ipcc 359 climate change simulations. Journal of Climate, 21(8), 1669–1679. 360 Yashayaev, I., & Clarke, A. (2006). Recent warming of the labrador sea. AZMP Bul-361 letin PMZA, 5, 12–20. 362 Yashayaev, I., & Loder, J. W. (2016). Recurrent replenishment of labrador sea wa-363 ter and associated decadal-scale variability. Journal of Geophysical Research: 364 Oceans, 121(11), 8095-8114. 365